Author's response to reviews

Title: Physiological responses to low-force work and psychosocial stress in women with chronic trapezius myalgia.

Authors:

Anna Sjörs (anna.sjors@liu.se)
Britt Larsson (britt.larsson@liu.se)
Joakim Dahlman (joakim.dahlman@liu.se)
Torbjörn Falkmer (torbjorn.falkmer@liu.se)
Björn Gerdle (bjorn.gerdle@liu.se)

Version: 3 Date: 3 June 2009

Author's response to reviews:

Response to the comments from Reviewer 2:

Correlations between muscle activity and pain for the CON are now addressed in the Results section.

The statement about causal relationships between pain and EMG activity has been clarified by adding the sentence: We do not know if the increased muscle activity is a cause or a consequence of pain, since it is unknown whether this was also the case before pain onset.

Formatting changes:

The required details for the references have been provided except for the reference in press. It has not yet been published.

Sincerely, Anna Sjörs